Author's response to reviews

Title: A young female with massive hemoptysis and deep venous thrombosis - Hughes Stovin syndrome: a case report

Authors:

Hamdan Al-Jahdali (jahdali@yahoo.com)

Version: 9 Date: 30 January 2010

Author's response to reviews: see over
30-01-2010

To: The BioMed Central Editorial Production Team

RE: Title: A young female with massive hemoptysis and deep venous thrombosis: HUGHES-STOVIN SYNDROME.
Journal: Journal of Medical Case Reports
MS : 1220936179247826

Dear Sir:

I have submitted the revised form of the above manuscript: The following is my responses to each concern:

Major revisions (we require the authors to make these changes)

**Query 1:** Manuscript information: please provide the ethnic origin of the patient.

**Answer:** Stated clearly in manuscript that the patient is Saudi young Female. I am not sure about meaning of this question.

**Query 2:** Conclusion: the manuscript must have a conclusions section provided in the main manuscript, please amend this accordingly.

**Answer 2:** Done. The following statement added as conclusion

"**Conclusion:**

This case report emphasizes the importance of Hughes-Stovin syndrome as a cause of hemoptysis secondary to pulmonary artery aneurysm in a young woman. Furthermore, it can increase awareness among physicians of the pulmonary manifestations of this Disease. Our patient was treated successfully with multi-modality treatment that includes surgery, steroids, and cytotoxic agents."

Sincerely
Dr. Hamdan AL-Jahdali
jahdali@yahoo.com